CS150 Newton 6.2. 1

Outline

m Last time:

Ü Quiz Review

m This lecture:

Ü Adders and subtracters : Half adder, full adder

Ü Adder organizations: ripple,carry look-ahead(CLA), carry

bypass

Ü Full adder as a full subtracter

Ü Sign-magnitude

Ü One’s & Two’s Complement

Ü BCD, Floating-point

Bits, Bytes, Nibbles and Words:

Some Definitions

m Data types which are ordered lists of binary digits often

have the following names:

Ü Bit: 1 digit

Ü Nibble: 4 digits

Ü Byte: 8 digits

Ü Word: The standard memory bus width in your architecture

(e.g. 16-bit, 32-bit, 64-bit words).

Addition

21

+ 6

10101

+ 110

1990

+ 215

Decimal Addition:

Binary Addition:

CS150 Newton 6.2. 4

Bit-Ordering, Byte-Ordering and the Endian